Author’s response to reviews

Title: Sustained serological and complete responses in HBeAg-Positive Patients treated with Peginterferon alfa-2b: a 6-Year Long-Term Follow-Up of a Multicenter, Randomized, Controlled Trial in China

Authors:

Jian Sun (sunjn09@126.com)
Huiguo Ding (dinghuiguo09@163.com)
Guofeng Chen (chengf10@yeah.net)
Guiqiang Wang (kingqiang06@163.com)
Lai Wei (weilaiwl09@163.com)
Jiming Zhang (zhangjm06@yeah.net)
Qing Xie (drxieqing5@126.com)
Mobin Wan (wanbomin13@126.com)
Hong Tang (tanghongc@yeahmca.cn)
Shijun Chen (cshijun17@yeah.net)
Zhiliang Gao (zhiliang10@mca21cn.cn)
Yuming Wang (kingym11@163.com)
Dazhi Zhang (zhdazhi15@126.com)
Wenxiang Huang (huangwenxiangwx@163.com)
Jifang Sheng (shengjif6@yeahmca.cn)
Qin Ning (ningqian06@163.com)
Dongliang Yang (dongliangy@mca21cn.cn)
Jian Lu (lujianlj09@126.com)
Chen Pan (chenpan16@yeah.net)
Yuxiu Yang (yangyx12@126.com)
Jue Wang (kingjue13@yeahmca.cn)
Chuanzhen Sun (sunchz5@mca21cn.cn)
Qixin Wang (qeexinw16@126.com)
Jinlin Hou (houjinlin08@mcatom.cn)

Version: 1 Date: 25 Jan 2018

Author’s response to reviews:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response: Done

2. Please add a statement on conflict of interests for all the other authors.

Response: Done

3. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

Response: Done

If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.

4. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: Done
5. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response: Done